Author's response to reviews

Title: A randomized, double blind, crossover, placebo-controlled clinical trial to assess the effects of a mouthwash containing chlorine dioxide on oral malodor

Authors:

- Kayoko Shinada (shinada.ohp@tmd.ac.jp)
- Masayuki Ueno (ueno.ohp@tmd.ac.jp)
- Chisato Konishi (mori.ohp@tmd.ac.jp)
- Sachiko Takehara (takehara.ohp@tmd.ac.jp)
- Sayaka Yokoyama (yokoyama.ohp@tmd.ac.jp)
- Yoko Kawaguchi (yoko.ohp@tmd.ac.jp)

Version: 8 Date: 5 December 2008

Author's response to reviews: see over
Dear the Trials Editorial Team

Thank you for the acceptance of our manuscript.

We tried to change the manuscript according as your requests.

Please find attached the changed manuscript.

If we have any problems about it, please contact me.

All authors have approved this revised submission.

Sincerely yours,

Kayoko Shinada DDS, PhD

Department of Oral Health Promotion, Graduate School,

Tokyo Medical and Dental University

1-5-45 Yushima, Bunkyo-ku, Tokyo 113-8549, Japan.

TEL: 81-3-5803-5476  FAX: 81-3-5803-0194